

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Holmlund, Bertil; Liu, Qian; Nordström Skans, Oskar

Working Paper Mind the gap?: Estimating the effects of postponing higher education

CESifo Working Paper, No. 1792

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Holmlund, Bertil; Liu, Qian; Nordström Skans, Oskar (2006) : Mind the gap?: Estimating the effects of postponing higher education, CESifo Working Paper, No. 1792, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: https://hdl.handle.net/10419/25837

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

MIND THE GAP? ESTIMATING THE EFFECTS OF POSTPONING HIGHER EDUCATION

BERTIL HOLMLUND QIAN LIU OSKAR NORDSTRÖM SKANS

CESIFO WORKING PAPER NO. 1792 CATEGORY 4: LABOUR MARKETS AUGUST 2006

An electronic version of the paper may be downloaded• from the SSRN website:www.SSRN.com• from the RePEc website:www.RePEc.org• from the CESifo website:www.CESifo-group.de

MIND THE GAP? ESTIMATING THE EFFECTS OF POSTPONING HIGHER EDUCATION

Abstract

This paper estimates the effects on earnings of "gap years" between high school and university enrollment. The effect is estimated by means of standard earnings functions augmented to account for gap years and a rich set of control variables using administrative Swedish data. We find that postponement of higher education is associated with a persistent and non-trivial earnings penalty. The main source of the persistent penalty appears to be the loss of work experience after studies. The reduction of lifetime earnings associated with two years postponement of higher education amounts to 40-50 percent of annual earnings at age 40.

JEL Code: I23, J24, J31.

Keywords: timing of education, schooling interruptions, returns to work experience.

Bertil Holmlund Department of Economics Uppsala University Box 513 75120 Uppsala Sweden Bertil.Holmlund@nek.uu.se

Qian Liu Department of Economics Uppsala University Box 513 75120 Uppsala Sweden Qian.Liu@nek.uu.se

Oskar Nordström Skans Institute for Labour Market Policy Evaluation (IFAU) Box 513 75120 Uppsala Sweden Oskar.Nordstrom_Skans@ifau.uu.se

This version: August 15, 2006

We thank seminar participants at Uppsala University as well as Iida Häkkinen, Hessel Oosterbeek and Björn Öckert for helpful comments. Financial support from the Swedish Council for Working Life and Social Research (FAS) is gratefully acknowledged.

1. Introduction

Conventional human capital theory predicts that investment in formal education occurs early in the life cycle. Formal education should precede work, a result implied by models where individuals attempt to maximize their wealth. By specializing in schooling early in life, the period over which the returns to investment can be reaped is extended. Indeed, this time sequence of schooling followed by work is also what we observe as the typical real-world pattern.

However, this time sequencing does not hold universally. Some young people prefer to postpone the start of their university education one or several years whereas others take a break in the middle of their education. In British vocabulary, the phenomenon of taking a year's break between high school graduation and university is known as doing a "gap year". Anecdotal evidence reported in the media suggests that gap years are common in the UK, and possibly on the rise in the US.¹ In Sweden, there also seems to be a popular perception that the prevalence of gap years is on the rise and the Danish "Welfare commission" reports that gap years in Denmark are frequent and amounting to several years on average (Velfærds-kommissionen, 2005).

Should the existence, and possibly increasing prevalence, of gap years be a matter of concern? Some have argued that gap years are socially costly as they presumably reduce the number of years of work after university graduation.² If the postponement of university education carries no positive productivity effects, this concern may be warranted. But there seems to be very little evidence about how one or two gap years after high school affect labor market outcomes after university graduation. The present study aims to fill some of this gap in knowledge. In particular, we will provide estimates of how gap years affect earnings after university. The policy relevance of the question is clear: if taking a year off is wasteful rather than productive, policies should make it more costly to postpone higher education.

A variety of reasons for gap years are conceivable. To fix ideas, we may distinguish between at least five types of gaps, viz. (i) gaps as investment in skills, (ii) gaps as leisure, (iii) gaps as

¹ The prevalence of school interruptions in the US is documented in Light (1995) and Monks (1997). Some data suggest that one-third of high school graduates interrupt their schooling before re-enrolling in college. In Britain, a thriving "gap year industry" has emerged which offers gap year ideas, gap year travel destinations, gap year jobs etcetera. A Google search on "gap year" returns over two million hits.

 $^{^{2}}$ See Landell et al (2000) for a discussion of the Swedish case and Velfærdskommissionen (2005) for an analysis of detours in the Danish education system.

waiting for better educational opportunities, (iv) gaps as learning about ones preferences and/or ability, and (v) military service.

Investment activities during gap years can take the form of work that is relevant for future studies and/or the career subsequent to university graduation. This includes behavior that improves work related skills, such as learning punctuality and good work habits. It could also involve the acquirement of social skills and language skills; indeed, anecdotal evidence on gap years often identifies "traveling abroad" as an important element of the gap period. Needless to say, the borderline between investment in skills and consumption of leisure is fuzzy when investment involves traveling abroad.

Some models of optimal investment in human capital with endogenous labor supply and perfect capital markets do allow for the possibility of bunching leisure early in life (Blinder and Weiss, 1976). This requires high "impatience", and in particular that the subjective rate of time preference is sufficiently large relative to the rate of interest. This may perhaps be seen as an exotic possibility. However, one can argue that the "capacity" for some forms of leisure consumption may be age-dependent. For example, the absence of family commitments around age 20 may enhance the range of feasible leisure consumption alternatives (long-distance traveling being one example).

A gap year may also sometimes be described as an "involuntary" pause that is the result of a failure to get accepted at the preferred education. The search for educational opportunities, like job search, involves applications, waiting for offers, and acceptance decisions if offers appear.³ Failure to get an offer, or rejection of offers, means in practice that the student has to wait until the next year's application round. The waiting period can be used in various ways, including further education so as to improve the chance of getting an acceptable offer next year. In Sweden, high school graduates are offered a second chance: they can take part in special courses so as to improve their grade points or they can take new and more advanced high school courses.

Finally, one can also imagine that a gap year might be a useful response to uncertainty about preferences and/or ability. If you can't decide what you want to do in life, why not postpone the

³ Comay et al (1973) formalize a sequential schooling decision problem with exogenous probabilities of advancing through a number of education levels. Heckman et al (2003, 2005) include elaborations and extensions of this approach. Altonji (1993) is a related contribution where education decisions under uncertainty are modeled.

decision until tomorrow, or next year? Taking a gap year might be individually (and possibly socially) beneficial to the extent that it reduces uncertainty about preferences and/or ability and thereby also the risk of choosing the wrong track.

Our aim in this paper is to offer evidence on how delayed transitions to higher education affect subsequent earnings and lifetime earnings. We exploit rich Swedish register data with complete coverage of the population of interest, i.e., individuals who have started a university education.⁴ These data allow us to retrieve information about direct and delayed transitions to higher education. We merge the data with other registers that contain information on earnings and employment as well as a host of other variables.

The results suggest that gap years are associated with lower future earnings. Individuals in their 30s earn less, the more gap years they have taken. The earnings penalty gradually diminishes over time and has vanished around age 40. This pattern is a robust feature of the data and persists after controls for a vast number of personal characteristics, including detailed information about levels and fields of education as well as measures of high school performance. The main source of the earnings penalty associated with gap years is that the returns to post-university work experience are higher than the returns to gap years. Postponement of university education reduces time for post-university investment in skills and therefore entails lower earnings subsequent to university graduation. The discounted present value of foregone lifetime earnings associated with two gap years amounts to around 40-50 percent of annual earnings at age 40.

The remainder of this paper is organized as follows. We begin in the next section with a brief review of related literature. Section 3 presents the data that we use and Section 4 gives a descriptive overview of how the timing of transitions to university education has evolved over recent decades. Section 5 discusses the empirical strategies and presents the results. Section 6 concludes.

2. Related Literature

Our paper is related to the vast literature on how the individual's work history affects subsequent earnings. Since Mincer (1974), it has become conventional to specify log earnings as a function of a quadratic in cumulative work experience (in addition to schooling and a few other

⁴ By 'university' we mean universities as well as university colleges.

variables). Although this specification has stood up reasonably well to some simple alternative functional forms,⁵ it is not well suited to deal with discontinuous labor force careers. The literature on the effects of (female) work interruptions, pioneered by Mincer and Polachek (1974), has seen a large number of specifications where the work history is characterized in substantial detail. A good example is Light and Ureta (1995), who include an array of variables that capture the timing and length of work spells over the career, a specification that is found to outperform Mincer's original specification.⁶ By and large, there is evidence that work interruptions are associated with wage penalties, but there is considerable variation in the magnitude of the estimated effects.

A smaller literature has focused on the returns to work while in school. Hotz et al (2002) is a recent contribution that also includes references to the previous literature on this topic. This literature has been mainly preoccupied with how work while in high school affects subsequent wages of (young) men. A number of studies based on US data have asserted that work while in school brings positive and long-lasting wage returns. One possible causal interpretation could be that early work experience produces useful human capital, such as good work habits etcetera. However, there is an issue as to whether the results are truly causal or reflect unobserved individual characteristics. Hotz et al (2002) find that previous estimates are dramatically diminished when controls for selection are introduced.

Most studies dealing with the earnings effects of combining education and work have dealt with high school education. There is at least one paper that focuses on how working while in university affects future labor market outcomes, namely Häkkinen (2006). She makes use of data from Finland and estimates a positive, albeit not long-lasting, effect of combining work and university studies.

It is often argued that the ability to learn is declining with age ("you can't teach an old dog new tricks"). It seems unlikely that this argument carries much weight as long as modest postponements of higher education are considered. In fact, there is at least some evidence that

⁵ See Heckman and Polachek (1974). Heckman et al (2003) offer a more critical assessment.

⁶ In addition to the references above, the literature on work interruptions and their effects on wages include, among many other contributions, Baum (2002), Corcoran and Duncan (1979), Mincer and Ofek (1982), Corcoran et al (1983) and Gronau (1988). Swedish evidence concerning the effects of work interruptions due to parental leave is given in Stafford and Sundström (1996) and Albrecht et al (1999). Nordström Skans and Lindqvist (2005) provide quasi-experimental estimates of the effects of career breaks by exploiting information from a Swedish labor market program that introduced subsidized work interruptions.

academic performance is increasing in age. Smith and Naylor (2001) examine the determinants of degree performance among undergraduates at UK universities in 1993. They find that older students generally perform better than younger ones, controlling for a rich set of covariates.

There is a very small US literature on the effects of schooling interruptions that is most closely related to what we do in the present paper. Griliches (1980) uses NLS Young Men data with information on schooling interruptions and estimates wage equations based on data from 1970. He finds no evidence that schooling interruptions have had any adverse effect on subsequent wage rates. Marcus (1984) uses the same data as Griliches but focuses on wage outcomes in 1973 and includes more elaborated specifications than those adopted by Griliches. The return to time spent as interrupter of schooling is found to be significantly lower than the return to potential work experience subsequent to completed schooling. Light (1995) uses longitudinal data from 1979 to 1989 based on NLSY, focusing on white males. The results vary by the individuals' completed education. Among individuals with 12 years of school, there are negligible wage differences between those with and without schooling interruptions. Among those with 16 years of schooling, however, there is evidence of a wage penalty associated with schooling interruptions. Finally, Monks (1997) extends Light's analysis by using NLSY from 1970 to 1993 and by including women and minorities in addition to white males. The main result is that those who complete college at a later age receive a smaller initial increase in the wage than those who have completed college earlier in life.

In summary, there is overwhelming evidence that life cycle events concerning work and education are key determinants of individual earnings. However, only a handful of studies have addressed how labor market outcomes are affected by the *timing* of work and higher education. The results from these studies are mixed and their general applicability is uncertain. Our study sheds new light on how the timing of education affects wages, earnings and employment. In all likelihood, our Swedish register data on education are less plagued by measurement errors than the survey data used by previous researchers.

3. The Data

3.1 Data Sources

We study individuals with completed university education using the Swedish IFAU-database which combines data from various population-wide registers constructed by Statistics Sweden.

We use register data on earnings, wages and employment as well as detailed data on university and pre-university education which we combine with census information on parents' education and socio-economic background.

Our data capture the year of each individual's first university enrollment from 1977 onwards and we have data on all high school graduations from 1985. We measure gap years in two ways. The first is the *True Gap*, denoted by *G* and defined as the time between the year of high school graduation (*YHSG*) and the year of first university enrollment (*YUE*), i.e., G = YUE - YHSG.⁷ This measure requires information about high school graduations and since these only are available from 1985, we can only use cohorts born in 1966 or later. We focus most of our analysis on studying true gaps. Our second measure is *Estimated Gap*, Ge = YUE - 19, where 19 corresponds to the normal age of high school graduation. The point of using this alternative measure is that we are able to use more cohorts (from 1958) since it does not rely on high school graduation information. Therefore, we can evaluate more long run effects when we use estimated gaps. In cases where we can compare the two measures, they are highly correlated and produce very similar results.⁸ We are interested in the effects of the timing of studies among university educated individuals, who started their university studies adjacent to, but not necessarily directly following, high school graduation. To that end we focus on individuals taking up to four gap-years between high school and university enrollment.

Our last year of data coverage is 2002 and since wee primarily study outcomes at age 30 or later we can only use cohorts born in 1972 or earlier. Naturally, when outcomes are measured at a higher age, this restriction is put further back. In our main analysis, we restrict the analysis to those who complete their university education within seven years of study in order to ensure that all included individuals have completed their education by the time they turn 30. Completion of studies is defined as graduating with a degree that (formally) corresponds to at least three years of study.

Our main outcome variables are annual labor earnings and wages, both derived from the employment registers (RAMS) in the IFAU data base. The outcomes are available between 1985 and 2002. The employment variables we use from this dataset capture annual labor earnings and

⁷ Multiple high school graduations are possible. However, over 95 percent of the graduates have just one graduation year, and we always use the first graduation year.

⁸ A linear regression of *G* against *Ge* yields a coefficient on *Ge* of 1.05 with R-square equal to 0.99.

employment status (as of November each year).⁹ We use the number of years an individual is employed to calculate work experience before, during and after university studies.

We construct a measure of wages as an alternative outcome variable. Data include information on the tax-records filed by each employer and from these records we can characterize each employer-employee relationship during a calendar year. We use two pieces of information, where the first piece contains the first and last month of an employment spell with an employer during the year. The second piece is information on the annual wage bill paid to the individual during this spell. We calculate wages by dividing the wage bill paid during an employment spell with the number of months the spell lasted. We keep spells with a wage that exceeds 75 percent of a minimum wage and that cover November.¹⁰ We also use at most one spell per individual (preference to the highest wage). Thus, we receive a wage measure that is corrected for the number of months of employment, but not for working hours. This procedure follows Nordström Skans et al (2006), who show that the derived wage distribution mimics the true wage distribution well.

In order to capture joint determinants of gaps and outcomes we rely on the inclusion of observed covariates. Our perhaps most important variable is a "field fixed effect" capturing the interaction between the formal length of the education (such as bachelor, master or doctoral) and direction of study for the degree.¹¹ This leaves us with around 450 different possible field dummies. This allows us to restrict the analysis to comparisons within professions such as "math/science teachers", "medical doctors specialized in radiology", and "bachelors of economics" instead of pooling all of these together. Furthermore, we use data from RAMS on year of birth, gender, county of residence the first time the individual is observed, and immigrant status. When studying true gaps, we also control for the grade point average (GPA) at high school¹² and the length of the high school education (2 or 3 years).¹³

⁹ Labor income includes all incomes paid by the employer that are subject to social security fees and payroll taxes.

¹⁰ The minimum wage is defined by wages for municipality-employed janitors. That all spells are required to cover November follows Statistics Sweden's definition.

¹¹ Measured at the most detailed level of the "SUN 2000"-coding system.

¹² Grades were at the time set on a scale from 1 to 5, and population GPA should have an overall normal distribution with mean 3 and standard deviation 1.

 $^{^{13}}$ During the time of study, Swedish high school education was divided into short (two years) mostly vocational programs and long (three years or – rarely – four years) academic programs. We include individuals with two or three year high school education.

Our final data source is the 1990 national census from which we derive information on parents' socioeconomic and educational backgrounds. There are 11 different socioeconomic groups capturing whether the parent is employee or self-employed, skilled or unskilled, manual or non-manual, working in manufacturing or in services etcetera. For parents' educational background, we have three categories, viz. less than high school, high school and university. In a sensitivity analysis we use sibling fixed effects defined from having the same biological mother. This analysis does however reduce the sample size dramatically.

Table 1 shows the years that define our sample. A more detailed description is provided in the appendix (Figure A1). Computation of estimated gap years requires only information on the year of university enrollment. The members of the first cohort we study are born 1958 and are thus estimated to graduate from high school in 1977. The last year of our observation window is 2002. Labor market outcomes in 2002 will thus involve persons aged 30 to 44, corresponding to birth cohorts from 1972 back to 1958.

Table 1. Y	Table 1. Years of importance for the data construction.							
1958	1966	1972	1977	1985	199			

1958	1966	1972	1977	1985	1990	2002
First included birth cohort. (Age 19 in 1977.)	First birth cohort with high school information. (Age 19 in	Last included cohort. (Age 30 in 2002.)	Start of register on first university enrollment	Start of register on high school information. Start of register	Year of measurement of family background.	Last year of measuring outcomes.
	1985.)			on outcomes.		

Table 2 shows how the resulting samples are affected by various restrictions. A large fraction of observations are lost by our exclusion of individuals with gaps greater than four years, especially for the age 35 sample. The reason for this restriction, which admittedly is somewhat arbitrary, is that we wish to focus on what may be deemed as "normal" gap periods, i.e., gaps that may typically be part of the young student's career plan. Adult education has become increasingly common in Sweden, but we conjecture that it is typically driven by other forces than those determining educational decisions among youths. As mentioned, we also require that the individuals have graduated within 7 years of study. This restriction, which allows us to analyze post university outcomes from age 30 and beyond, implies a further reduction of the sample at age 35.¹⁴ We examine the robustness of our results to both the length of the gaps and to the

¹⁴ A detailed description of pathways through Swedish university education is given in Statistics Sweden (2003).

restriction on graduation within 7 years and show that none of these restrictions are driving our results.

An additional restriction kicks in when we run earnings regressions with log earnings as the dependent variable. However, the overwhelming majority the university graduates that we consider have positive earnings so this restriction does not cause much further reduction in the number of observations. When studying wages, samples are reduced further (not shown in Table 2).

	Univ. educ. individuals at age of outcome (30 or 35)	Information about year of first registration	High school information (only <i>G</i>)	High school graduation at age 20 the latest (only <i>G</i>)	Gaps between 0 and 4 years	3-7 years between first registration and univ. graduation	Resident and positive earnings at age of outcome
<i>G</i> at age 30 Cohorts: 1966- 1972	133,431	121,189	108,998	106,337	91,146	73,558	70,899
G at age 35 Cohorts: 1966- 1967	44,535	38,676	30,850	30,170	23,157	15,812	15,209
<i>Ge</i> at age 30 Cohorts: 1958- 1972	233,552	204,988			169,369	136,467	131,281
<i>Ge</i> at age 35 Cohorts: 1958- 1967	172,955	142,688			103,092	71,690	68,833

Table 2. Number of observations for subgroups and after imposed restrictions.

Note: High school information is available from 1985. *G* stands for "True gap" between high school and university, *Ge* stands for "Estimated gap" which assumes age 19 as high school graduation age.

4. University Enrollment and Gap Years

University enrollment has risen substantially in Sweden since the early 1990s. The number of university entrants amounted to 51 000 in the academic year 1990/91 and had risen to 66 000 in 1995/96; by 2003/04, the number had risen further to 83 000 persons.¹⁵ Some of this development is accounted for by an increasing number of foreign students, but the major factor is a trend increase in transition rates from Swedish high schools to universities. We first examine

¹⁵ These numbers (from Statistics Sweden) refer to persons who are registered for the first time at any Swedish university or university college.

how direct and delayed transition rates from high school to university have evolved over time. This is followed by an inspection of the distribution of gap years among university entrants.

4.1 Direct and Delayed Transitions to University

High school graduation typically occurs in May or June and university enrollment can take place earliest during fall or spring the following academic year. We define a "direct transition" as a transition that takes place within one year; delayed transitions are thus transitions that take place after two or more years.¹⁶ Direct transitions correspond to true gaps of zero or one years. Figure 1 shows the development since the late 1980s.



Figure 1. Direct and delayed transitions to university, 1989-2003, percent.

Notes: Direct transition is defined as a transition within one year after high school graduation. Graduation during 1988/89 is labeled 89, 1989/90 is labeled 90, etc. The transition rate for 1994/95 is affected by organizational changes of high school education.

Source: Statistics Sweden (UF 36 SM 0201, UF 36 SM 0401, SCB:s website).

There are marked trend increases in direct as well as two-year transition rates. The two-year transition rate has doubled since the late 1980s and by the turn of the century it has reached the same level as the direct transition rate. The fraction of a cohort of high school graduates who have enrolled at the university after five years was 30 percent in the late 1980s and had increased

¹⁶ These transition rates are published by Statistics Sweden (Statistical Reports, series on the transition from upper secondary school to higher education) and originate from the same primary data sources as we use in most of the paper.

to over 50 percent by the late 1990s. The data also reveal marked gender differences in transition rates (not shown), with women having higher direct as well as delayed transition rates. These gender gaps in transition rates have widened since the late 1980s.

4.2 The Evolution of Gap Years

Let us now turn to the evolution of gap years in our (unrestricted) data set. Figure 2 shows the distribution of estimated gap years among university entrants over the period 1977-2002. There is a marked increase in the 0-1 category in the early 1990s, a development that is reversed during the second half of the decade. This pattern is broadly consistent with the evolution of direct transition rates over the period; cf. Figure 1. Roughly speaking, the sharp increase in enrollment is largely driven by direct transitions in the early 1990s, and to a more significant degree by delayed transitions in the late 1990s.



Figure 2. The distribution of estimated gap years among university entrants 1977-2002, percent.

In Figures 3 and 4, we plot the mean and median estimated gap by first year of university enrollment over the period 1977-2002.¹⁷ The mean estimated gap over this period is around 6 years, whereas the median gap is 3 years. The large difference between the mean and the median reflects a highly skewed distribution. We also notice that women generally have longer gaps than men.

¹⁷ We do not plot true gaps due to the lack of information on year of high school graduation before 1985.

Figure 3. The evolution of male estimated gap years, mean (solid) and median (dashed), 1977-2002.



Figure 4. The evolution of female estimated gap years, mean (solid) and median (dashed), 1977-2002.



All in all, it is clear that gap years between high school and university are common. However, the data do not speak loudly in favor of the view that the prevalence of gap years has increased over time. As Figure 2 makes clear, the fraction of university entrants with long gaps has fallen since the late 1970s. Moreover, the fraction with zero or short gaps has increased over the 1990s.

4.3 Who are Doing Gap Years?

We now turn to descriptive regressions aimed at documenting individual characteristics that correlate with gap behavior. We estimate a linear probability model for taking more than one gap year, with and without controls for field of education (FE). Each FE-category is a unique combination of the level (undergraduate or graduate) and the field of education. We also show results for a linear model with G as dependent variable, also with and without FE-controls. The results are shown in Table 3.

A brief summary of the results runs as follows. Women are less likely than men to take gap years, a pattern that differs from what we have seen in the raw data. Immigrants from non-Nordic countries are also less likely to take gap years and the same holds for students with higher GPA's. Students with two-year high school education are more prone (relative to those with three years) to take gap years; this is as should be expected since they may need a gap year to qualify for some university educations. The effects of parents' education are not strong and systematic.¹⁸ Moreover, the results do not seem to be much affected by whether or not the FE-controls are included.

One message to take away from Table 3 is that high school performance matters: the stronger performance in high school (as measured by GPA), the less likely to take gap years. This is as expected, since strong high school credentials imply generally higher chances of being accepted in any university education. One motive for gap years – waiting for better educational opportunities – is largely irrelevant for the top students.

An important issue that arises when attempting to estimate the effects of gap years is whether those are correlated with omitted individual characteristics ("ability"). If high ability students are much more likely to eschew gap years, the estimated gap effects may turn negative simply for omitted variable reasons.¹⁹ Our data do not make it possible to fully resolve this issue, but we will argue that the problem is unlikely to be severe since the results are robust to changes in observable characteristics. For example, the results remain essentially intact irrespectively of whether or not high school information is included.

¹⁸ The results are virtually identical if parents' socioeconomic class is excluded.

¹⁹ Interestingly, however, Griliches (1980) reports that individuals with schooling interruptions in his US sample have *higher* IQ's and come from more affluent backgrounds.

Table 3. Who are doing gap years?

	Linear proba Prob (bility model: (G>1)	Linear model with G as dependent variable		
Female	-0.035	-0.073	-0.238	-0.276	
	(0.003)**	(0.003)**	(0.009)**	(0.010)**	
Born in the Nordic Countries	-0.019	-0.024	0.01	0.001	
	(0.019)	(0.019)	(0.058)	(0.052)	
Born in Western Europe or	-0.054	-0.052	-0.192	-0.191	
North America	(0.025)*	(0.023)*	(0.066)**	(0.064)**	
Other Immigrants	-0.086	-0.082	-0.301	-0.287	
	(0.013)**	(0.012)**	(0.036)**	(0.034)**	
2-year High School	0.050	0.052	0.342	0.367	
	(0.002)**	(0.003)**	(0.009)**	(0.008)**	
GPA	-0.188	-0.179	-0.857	-0.832	
	(0.003)**	(0.003)**	(0.008)**	(0.009)**	
Father's Education	-0.007	-0.005	-0.029	-0.032	
Less than High School	(0.005)	(0.005)	(0.014)*	(0.013)*	
University	-0.004	-0.005	-0.082	-0.088	
	(0.005)	(0.005)	(0.015)**	(0.014)**	
Mother's Education Less than High School	0.001	0.001	-0.012	-0.015	
University	0.013	0.009	-0.012	-0.024	
	(0.005)*	(0.005)	(0.015)	(0.014)	
Observations	73,558	73,558	73,558	73,558	
R2	0.07	0.08	0.18	0.18	
FE / # FE	No	452	No	452	

Notes: Individuals born between 1966 and 1972 are included in the sample. The regressions also include 6 birth year dummies, 10 dummies respectively for the fathers' and mothers' socioeconomic classes and 24 dummies for county of residence at age 16. R2 with FE refers to R2 within. Robust standard errors in the parentheses.

* significant at 5%; ** significant at 1%.

5. Empirical Analysis of Gaps and Earnings

We now turn to an empirical analysis of gaps, wages and labor income. We are ultimately interested in how the timing of university enrollment affects lifetime earnings. There is a presumption that postponement of enrollment is bound to reduce lifetime earnings since it most likely reduces time spent in the labor force subsequent to university graduation. However, the validity of the presumption depends, inter alia, on the extent to which gap years involve productive labor market activities; if they do, the associated frontloading of earnings may conceivably increase the present discounted value of lifetime earnings. The earnings' effects of gap years depend also on the extent to which they bring about on-the-job training that carries a return in terms of earnings subsequent to university graduation. Finally, the earnings' effects of gap years hinge crucially on the value of post-university labor market experience relative to the value of the gap experience.

5.1 Gap Years and Subsequent Earnings

Estimation of the returns to gap years raises all the usual problems encountered in the huge literature on returns to schooling (and the somewhat smaller literature on returns to experience). Griliches (1977) contains an early discussion of some of the key issues involved; Card (1999) provides a recent survey. One potential problem is that gaps, like levels of schooling, are endogenously chosen by individuals, at least to some degree. A related problem is caused by the presence of omitted variables that affect education decisions as well as earnings. Measurement errors in schooling and work experience raise additional problems (although our data on education should be of high quality). However, it is noteworthy that a central message from the literature on the returns to schooling is that "the average (or average marginal) return to education fit by OLS". (Card, 1999, p 1802.) Our analysis relies on OLS estimates on cross section data, although with unusually detailed control variables for the field of education.

Let $y_{i,j}$ denote individual *i*'s annual earnings, where subscript *j* denotes the age group. We estimate cross-sectional earnings equations at different ages, explaining earnings by a set of covariates (*Z*) and measures of gap years:

(1)
$$\ln y_{i,j} = \alpha_j + Z_{i,j}\beta_j + \gamma_j G_{i,j} + \varepsilon_{i,j}$$

The set of control variables includes a gender dummy,²⁰ dummies for immigrant status, a dummy variable of having two-year high school (instead of three years), GPA score, 10 dummy variables for father's (mother's) socioeconomic group, father's (mother's) educational background, 24 dummies for county of residence at age 16, and dummy variables for year of birth. Most of the specifications also control for field of education. The specification of the equation for wages $(w_{i,i})$ is analogous.

We start our analysis at age 30, the first year when all included individuals have graduated from university. This is also the year when we have the most observations since we are able to include the cohorts 1966-1972 (when studying true gaps). Table 4 displays the estimates of the gap year effects. We start by estimating a model without the field of education effects (column 1). The estimates clearly suggest a negative effect of taking a gap year. We add controls for field and duration (of university education) in columns 2 to 3. The effects are not influenced much by these variables. The linear estimate implies that the earnings penalty is around 3 percent per gap year. In column (4) we relax the linear restriction for the gap effect and the evidence suggests a monotonic relationship between gap length and the earnings penalty. In column (5) we find a similar pattern for wages as we did for earnings. Whereas the earnings effect of an additional gap year appears to be increasing with the number of gap years, the wage effect appears almost strictly linear. The effects on employment probabilities are negative but less clearly monotonic (column 6).

The estimates in Table 4 imply that individuals with the same actual length, formal level and field of education earn less at age 30 if they had started their university studies later. Given that the durations (and age) are the same, potential experience is also equal, yet their earnings differ systematically. This suggests that the timing of education is indeed important, at least early in the post-education career.

²⁰ The results for men and women are similar and we have therefore pooled the data across gender.

	Depen	dent variable:	rnings	Dep. var. log wage	Dep. var. Prob(employment)	
	(1)	(2)	(3)	(4)	(5)	(6)
G	-0.036 (0.003)**	-0.024 (0.003)**	-0.029 (0.003)**			
Duration of studies			-0.034 (0.003)**	-0.034 (0.003)**	-0.023 (0.001)**	-0.012 (0.001)**
<i>G</i> =1				-0.016 (0.009)	-0.021 (0.004)**	-0.005 (0.003)*
G=2				-0.034 (0.009)**	-0.040 (0.004)**	-0.004 (0.003)
G=3				-0.071 (0.011)**	-0.066 (0.004)**	-0.008 (0.003)**
<i>G</i> =4				-0.130 (0.013)**	-0.093 (0.005)**	-0.016 (0.004)**
Observations R2 FE / # FE	70,899 0.12 No	70,899 0.05 450	70,899 0.05 450	70,899 0.05 450	57,900 0.16 427	73,558 0.01 452

Table 4. Gaps and labor market outcomes at age 30, cohorts born 1967-1972.

Note: Individuals born between 1966 and 1972 are included in the sample. The regressions include a gender dummy, dummies for immigrant status (Swedish born, Nordic, Western or Other), high school GPA, length of high school education, 6 birth year dummies, 10 dummies respectively for the fathers' and mothers' socioeconomic classes, 3 for the mothers' and fathers' education and 24 dummies for county of residence at age 16 (or in 1985, whichever comes first). R2 with FE refers to R2 within. Robust standard errors in the parentheses. * significant at 5%; ** significant at 1%.

Robustness

In Table 5 we show estimates when outcomes are measured at age 35. We first use the same basic restrictions, meaning that the individuals should all have completed their university education at age 30. Thus, they all have had at least five years to establish themselves on the labor market. In order to study the outcomes at age 35 we need to focus on the cohorts born in 1966 or 1967 so the sample sizes are reduced. Therefore, and to save space, we focus on the linear effect of gap years. We see evidence of some convergence over time (the earnings effect is -0.021 at age 35 which should be compared to -0.029 at age 30 in Table 4), but the effects are still significant and negative for both earnings and wages at age 35.²¹

²¹ The same pattern of gradual convergence is also evident when following the cohorts 1966-67 over time.

In Table 5 we also relax the restriction of 1-4 years of gaps without much change in estimates, neither for earnings nor for wages. We also show estimates where we remove the restriction of graduating within 7 years, again without much change in results.²² We perform both of these robustness checks for the age 35 model since the restrictions have a larger impact at higher ages (there is more possible variation in both duration and gaps; see also Table 2). The fact that the results are robust in these dimensions suggests that our conclusions are not driven by the sample creation strategy.

		Earnings		Wages			
	(1)	(2)	(3)	(4)	(5)	(6)	
G	-0.021	-0.023	-0.026	-0.020	-0.018	-0.019	
	(0.006)**	(0.003)**	(0.003)**	(0.003)**	(0.001)**	(0.001)**	
Duration of studies	-0.034 (0.007)**	-0.024 (0.006)**	-0.021 (0.003)**	-0.023 (0.003)**	-0.019 (0.003)**	-0.016 (0.001)**	
Restrictions							
Gaps	0–4	All	All	0–4	All	All	
Duration	3–7	3–7	All	3–7	3–7	All	
Observations	15,209	20,321	28,398	13,060	16,974	23,411	
R2	0.08	0.07	0.06	0.13	0.12	0.11	
# FE	181	406	506	174	380	480	

Table 5. Sensitivity checks at age 35 – models with and without restrictions on Gaps and Durations.

Note: The regressions include a gender dummy, dummies for immigrant status (Swedish born, Nordic, Western or Other), high school GPA, length of high school education, 6 birth year dummies, 10 dummies respectively for the fathers' and mothers' socioeconomic classes, 3 for the mothers' and fathers' education and 24 dummies for county of residence at age 16 (or in 1985, whichever comes first). Education information is taken at age 30 for columns (1) and (4) for consistency with Table 4 and at age 35 for columns (2), (3), (5) and (6). R2 refers to R2 within. Robust standard errors in the parentheses. * significant at 5%; ** significant at 1%.

It should be noted that our identification relies entirely on observed characteristics for a causal interpretation of the estimates. The reason is that, as when studying the returns to the level of education, good instruments are extremely hard to come by. Furthermore, it is even more difficult to think of instruments that affect the *timing* of education and not the probability of actually pursuing higher education, which will cause problems if there are heterogeneous returns to education. In passing, it is worth noting that the converse also holds even though it has been

 $^{^{22}}$ Since the Swedish university system allows individuals to register for single (sometimes part-time) courses and since we are unable to separate "real" studies from such occasional part-time courses for those that do not complete a degree, we retain the restriction of graduation throughout. This also allows us to control for the exact type of education.

largely ignored in the literature – any instrument that affects participation is also likely to affect the timing, which may bias IV-based estimates of the returns to education.

With these caveats in mind we have explored three potential instruments that mainly should affect the timing although we can not exclude the possibility that the selection is affected as well. The three instruments are based on the timing of military service enrollment, the local labor market conditions at the time of high school graduation and an admissions reform. None of these attempts proved successful, however; the first stage estimates were far too imprecise to be informative.

Our most promising candidate instrument was a university enrollment reform performed in 1981-82 (see Högskoleverket, 1997). The explicit purpose of the reform was to reduce the number of gap years and it had several parts leading to this end. Most importantly, one third of slots were allocated specifically to those with less than three gap years, the possibilities of using work experience as a merit was reduced, and the possibility of using activities in non-governmental organizations as a merit was removed. The reform appears to have had an impact on the transition rates to university; the mean estimated gap (in the interval 0-4) dropped from 1.87 for the cohort born in 1961 to 1.62 for the cohort born in 1964 (the pattern is also visible in Figure 2 above). In order to use the reform as an instrument we adopted a regression discontinuity design. The instrument was defined as being born in cohorts 1963 or after and we assume that the underlying time and cohort pattern in real earnings follow a quadratic functional form (instead of using cohort dummies as previously). Under this assumption, cohorts born after 1963 should have shorter gaps but not different earnings for other reasons conditional on the quadratic term. The IV-estimates were considerably larger in absolute values (-0.085) than the OLS-estimates, but quite imprecise (the standard errors were 0.06). A Hausman-test could not reject the consistency of the OLS estimates.

Given that none of our candidate instruments provided enough power for identification, and the intrinsic difficulties of finding truly reliable instruments for the timing of education, our overall conclusion is that we are restricted to relying on observable characteristics for identification. We made some further robustness checks by including more covariates in the model. Estimates are provided in Table A3 in the appendix. First, we used sibling fixed effects (based on the identity of the biological mother) instead of the observed family background variables. Second, we extended our model by interacting GPA with *both* the field of university education *and* the field

19

of high school education (60 categories) simultaneously. The results are robust to both of these variations and we interpret this as evidence for the model being correctly specified both when it comes to controlling for family background, and when it comes to controlling for educational opportunities and (pre-university) scholastic abilities, lending support to our causal interpretation of the estimates.

5.2 Gap-Effects over the Lifecycle

So far we have restricted the analysis to wages and earnings at ages 30 and 35. We now extend the focus to a large part of the life cycle by estimating the returns to gap years at each age where observations are available. That is, we run separate regressions based on eq. (1), letting age vary from 20 to 44.²³ By doing so, we can examine whether the convergence patterns discussed above are visible in the data. In order to include all these age groups, we need to use estimated gaps, i.e., imputed gaps based on the normal age at high school graduation. Although this forces us to drop high school information from the model, we can use all cohorts from 1958 and onwards and thus capture 25 years of the life cycle.²⁴ We focus on earnings since we will use the same set-up to calculate the effects on real life time earnings in the following section.

Figure 5 displays the time pattern of gap estimates based on 25 separate regressions with log annual earnings as dependent variable. The patterns for men and women are very similar and the figure shows results from regressions where the data on men and women are pooled (and gender is accounted for by a dummy). In the early 20s, an additional gap year is associated with around 20 percent higher earnings. This is followed by a substantial dip in earnings around age 25, i.e., during the period of university enrollment when labor market activity is limited. After graduation there is a period of fairly slow catch up. The estimated effect at age 35 is 2.1 percent, exactly as in Table 5, where true gaps are used. Full convergence is eventually achieved in the early 40s; the estimated gap effects turn insignificant at age 42.

 $^{^{23}}$ Each regression is thus based on a different set of birth cohorts but the patterns are very similar between cohorts (see Figure A2 in the appendix). Controls for birth cohorts are included in each regression, as well as controls for field of education and other covariates discussed above (except high school information).

²⁴ We have also performed these exercises using the data on true gaps and with controls for GPA from high school. This can only be done for the age groups 20 to 36. The results are very similar to those displayed for estimated gaps.



Figure 5. Estimated returns to gap years by age.

Note: The estimated gap effects on log annual earnings for ages 20-41 are all significantly different from zero, with absolute *t*-values ranging from 2.8 (at age 41) to 158.5 (at age 25). The shaded area highlights the three insignificant estimates (ages 42-44).

5.3 Lifetime Earnings and the Costs of Gap Years

We will now illustrate the implications for lifetime earnings of the patterns shown in Figure 5. Note that an effect on log earnings matters more for real lifetime earnings if it occurs when real earnings are high. We therefore need to transform the estimated effects on log earnings into effects on real earnings. Ideally we would do this by using the age-earnings profile for a single cohort. However, since our data does not contain such information, we choose to focus on the cohort born 1965 where earnings information is available for ages 20-37.²⁵ For this cohort and this age interval, we compute real earnings by deflating by the consumer price index. Real earnings during ages 38-44 for the 1965 cohort are imputed by using information on the evolution of real earnings for the 1958 cohort over the 38-44 age interval. The resulting series is thus intended to capture real earnings for the 1965 cohort from age 20 and up to age 44, remembering that the seven last years are imputed from data pertaining to the 1958 cohort.²⁶ We

²⁵ Recall that we have information on at most 25 age groups (age 20-44), and 15 cohorts (born 1958-72). Observations on earnings are available for the years 1985-2002, which means that we do not have information on earnings for all 25 age groups for any single cohort.

 $^{^{26}}$ The earnings profile is "standardized" in the sense that it is based on estimated earnings equations of the form given by (1) above.

assume that gaps have no effects beyond age 44, which is consistent with our estimates presented above.

Equipped with a "representative" earnings profile we can translate gap effects on log earnings into gap effects on real earnings in units of SEK. Let $\ln y_{r,j}$ denote real log earnings for age *j* associated with our representative cohort, assuming no gaps. The corresponding profile with *Ge*=1 is then given as $\ln y_{r,j}^G = \ln y_{r,j} + \gamma_j$, where γ_j is the estimated gap parameter and $j \in \{20,...,44\}$ The key variable is the absolute difference between earnings associated with positive gaps and earnings associated with zero gaps, i.e.,

(2)
$$\Delta y_{r,j}^G = \exp(\ln y_{r,j} + \gamma_j \cdot Ge) - \exp(\ln y_{r,j})$$

where $Ge \in \{1, 2, 3, 4\}$. We have computed discounted present values of (5) for the 25 year period corresponding to ages 20-44, assuming a 4 percent discount rate and no earnings penalty after age 44. The results are given in Table 6.

		Men	Women			
	SEK	SEK Fraction of annual earnings at age 40		Fraction of annual earnings at age 40		
Ge=1	- 106 700	-0.21	- 92 100	-0.31		
Ge=2	- 187 800	-0.37	- 157 900	-0.53		
Ge=3	- 246 800	-0.49	- 200 600	-0.67		
Ge=4	- 286 100	-0.57	- 221 700	-0.74		

Table 6. The effects of gap years on the present value of lifetime earnings.

Notes: The discount rate is set to 4 percent. It is assumed that no gap effects on earnings persist after age 44. The calculations pertain to the cohort born 1965 as described in the text. The returns to gap years are estimated separately for men and women.

The results suggest indicate non-trivial effects on lifetime earnings. Two gap years carry a lifetime loss in earnings that amount to 37 percent of annual earnings at age 40 for men; for women, the corresponding loss amounts to 53 percent. The reason that relative costs are larger for women is mainly that they have substantially lower earnings than men at age 40, partly reflecting much higher incidence of part time work. In absolute terms, the present value losses of

gap years are somewhat higher for men. For Ge=2, the present value loss (in 2002 prices) amounts to around 188 000 SEK for men and 158 000 SEK for women.²⁷

The private costs of gap years will generally differ from those reported in Table 6 for mainly two reasons, viz. taxes and stipends. Progressive taxes reduce the net penalty of taking gap years analogous to the way they reduce the returns to higher education in general. Stipends, including grants and subsidized loans, may increase the incentives to study now rather than later, all else equal; postponement of studies involves also postponement of stipends and the more generous those are, the more costly the delay. However, the presence of stipends may also affect occupational choice by increasing the attractiveness of educations that have positive consumption values but perhaps modest effects on earnings.

5.4 Gaps and the Returns to Work Experience

How can we understand the persistent effects of gap years on earnings? Notice that we compare individuals with the same field, formal level and actual length of education (which means that they have the same amount of potential work experience), yet we find persistent effects on earnings even in the mid-30s. One possible explanation for this result is that gap years reduce the total amount of work experience and thereby the amount of human capital acquired on the job; another possibility is that gaps affect the timing of work experience (or "inactivity") in a way that is detrimental for earnings. To shed light on these issues, we proceed by estimating "time use equations" where individual time allocation after high school graduation is explained by the number of (true) gap years in addition to the covariates previously used. In a subsequent step we estimate earnings equations where our measures of time use are included as regressors. By doing so we can gain some understanding of how gap years affect earnings through the effects on time use after high school.

We focus on time use between high school graduation and age 35. This time span (*T*) is treated as fixed and can be allocated to work experience (*X*) or "inactivity" (*I*), where the latter is defined as $I \equiv T - X$.²⁸ Work experience is measured by using information about whether the individual is employed or not in November each year. This measure is thus binary for each year

²⁷ The exchange rates for July 2006 are 1 USD = 7.44 SEK, 1 EURO = 9.40 SEK.

²⁸ We control for age at high school graduation in the estimations.

but varies (in unit steps) depending on the number of years with work experience.²⁹ Moreover, we distinguish between work experience before, during and after university studies. Since inactivity by construction is the mirror image of work experience we thus get three inactivity variables (before, during, after) corresponding to the three experience variables.

The results are presented in Table 7. We see that one additional gap year is associated with 0.56 additional years of work before university, 0.05 additional years of work during university, and 0.75 fewer years of work after university. It is also associated with 0.44 years of extra inactivity before university, 0.19 fewer years of schooling without work, and 0.12 fewer years of inactivity after university. This sums, of course, to one year before start of university, and minus one year after start of university. Total work experience is reduced by 0.14 years. Since time is fixed it follows that total inactivity is increased by the same amount.

Table 7. Estimated effects of a gap ye	ar on activities betwee	en high school graduation and ag	ge 35,
cohorts born 1966-67.			

	Experience (X)	Inactivity (T-X)	Total (T)
	(1)	(2)	(3)
Before	0.560	0.440	1
	(0.004)**	(0.004)**	1
During	0.052	-0.185	-0.134
-	(0.010)**	(0.010)**	(0.007)**
After	-0.749	-0.117	-0.866
	(0.014)**	(0.012)**	(0.007)**
Total	-0.137	0.137	0
	(0.017)**	(0.017)**	0

Note: Inactivity is defined as not having worked. The model is estimated for 35 year olds using true gaps. The regressions include a gender dummy, dummies for immigrant status (Swedish born, Nordic, Western or Other), high school GPA, length of high school education, age at high school graduation, 6 birth year dummies, 10 dummies respectively for the fathers' and mothers' socioeconomic classes, 3 for the mothers' and fathers' education and 24 dummies for county of residence at age 16 (or in 1985, whichever comes first). Robust standard errors in the parentheses. * significant at 5%; ** significant at 1%.

So how can we make use of this information in order to understand the effects of a gap year? First of all we can note that a gap year *reduces* total work experience (thus increasing inactivity). Secondly, it places more of the experience *before* the studies, potentially reducing the value of the work experience. And thirdly, it changes the *timing of inactivity* by increasing inactivity before the studies and reducing it afterwards. To provide an interpretation of this pattern, we will explicitly incorporate actual work experience into our model. We allow for a quadratic of work

²⁹ Work experience is measured as the number of years with at least four hours of work performed in November. We have estimated models with stricter requirements, but the estimates are quantitatively similar. Results are available upon request.

experience after university. We choose wages as the outcome variable since the returns to experience typically are discussed in those terms; results for annual earnings are available on request. The results are displayed in Table 8.

	"Reduced form"	Years of experience	Timing o	f experience	Experience and inactivity
	(1)	(2)	(3)	(4)	(5)
Gap year	-0.017 (0.003)**				
Total experience		0.026 (0.002)**			
Exp. before			0.010 (0.004)*	0.008 (0.004)	-0.026 (0.004)**
Exp. during			0.016	0.015	-0.020
Exp. after			0.035	0.058	Ref.
(Exp. after)^2/100			(0.002)**	-0.140 (0.055)*	
Inactivity before					-0.024
Inactivity during					-0.036
Inactivity after					-0.038
Observations	13,077	13,077	13,077	13,077	13,077
R2	0.12	0.14	0.14	0.14	0.14
FE	174	174	174	174	174

Table 8. Estimated effects of different activities on wages at age 35.

Note: Inactivity is defined as not having worked. The model is estimated for 35 year olds using true gaps (Column 1). The regressions include a gender dummy, dummies for immigrant status (Swedish born, Nordic, Western or Other), high school GPA, length of high school education, age at high school graduation, 6 birth year dummies, 10 dummies respectively for the fathers' and mothers' socioeconomic classes, 3 for the mothers' and fathers' education and 24 dummies for county of residence at age 16 (or in 1985, whichever comes first). R2 refers to R2 within. Robust standard errors in the parentheses. * significant at 5%; ** significant at 1%.

The estimate in the first column, which can be loosely thought of as a reduced form, implies that an additional gap year results in a wage reduction by 1.7 percent at age 35. An increase in years of work experience is associated with a wage increase by 2.6 percent, as shown in the second column. The third and fourth columns show that the timing of work experience also matters; the returns to work experience after graduation is much higher than the returns to work before or during the studies. Finally, we notice from the fifth column that the timing of inactivity also plays a role. Experience after university is chosen as the omitted time use category so an increase in inactivity, holding experience before and during university constant, must be accompanied by a decrease in experience after university.³⁰ The estimates in the fifth column thus imply that a reduction in experience after university is more costly if it is driven by inactivity after university than by inactivity before university.

We now make use of the estimates in Table 7 and Table 8 to shed light on how gap years affect wages by influencing total work experience, the timing of work experience, and the timing of inactivity. If total experience is all that matters, we would get an expected effect of one more gap year by combining results from column (1) in Table 7 and column (2) in Table 8. In particular, since a gap year reduces total work experience we obtain the *years of experience effect* as the product of the returns to experience (0.026) and the decline in experience implied by a gap year (0.137). This effect amounts to a wage reduction of 0.4 percent.³¹ Thus, the effect on total experience can only account for a small fraction of the total effect.

Let us next recognize the fact that a gap year affects not only the total amount of work experience but also its timing. We combine the estimates of experience effects before, during and after studies from column (2) in Table 7 and the estimates in Column (3) of Table 8 and obtain a *timing of experience effect*. This effect – which incorporates the effect on the amount of experience as well as its timing – amounts to a wage reduction of 2.0 percent.³² Thus, the effect of timing in itself has a larger effect (-1.6 percent) than the effect through the amount of experience. Note that convergence over time can be understood by the negative quadratic term of post-university experience shown in Column (4).

Since the gap years also affect the timing of inactivity, we can use the "full" model of Column (5) of Table 8 with all the information in Table 7 to get a sense of how wages are affected by changes in the timing and amounts of both experience and inactivity. This *timing of experience and inactivity effect* implies a wage reduction of 1.6 percent.³³ Note that this is not far from the estimated "reduced form" effect of a gap year of -1.7 percent, as given in the first column of Table 8.

³⁰ The six time use categories sum to a fixed number and can thus not all be included in the regression.

³¹ This calculation as well as the subsequent ones in this section makes use of estimates with higher precision (more decimals) than those shown in Table 7 and Table 8.

 $^{^{32}}$ The relevant calculations are as follows: $0.560*0.010+0.052*0.016-0.749*0.035 \approx -2.0/100$.

³³ This is obtained as follows: $-0.560*0.026-0.052*0.020-0.440*0.024+0.185*0.036+0.117*0.038 \approx -1.6/100$.

The estimates, taken at face values, point to an interesting pattern. First, *total experience* matters since a gap year reduces the total amount of time spent working. Second, the *timing of experience* matters a lot: recall that a gap year substantially reduces experience after university and that the returns to experience after university are much higher than the returns to other forms of experience. Third, the *timing of inactivity* matters as well. Inactivity before studies is much less harmful than inactivity after education. This last estimate is perhaps somewhat unusual but given that that we know that work interruptions reduce future wages above what can be explained by lost experience it is perhaps not so strange that the timing of inactivity matters.

6. Concluding Remarks

The transition from high school to higher education typically involves one or several gap years. Among Swedish university entrants around the turn of the century, some 25 percent had taken two to four gap years, and around 40 percent had more than five gap years. Scattered evidence from other countries suggests that the phenomenon of gap years is not unique to Sweden.

The paper has examined how postponement of higher education affects earnings subsequent to the completion of the studies. We have focused on "short" delays, i.e., delays up to four years but our estimates are robust both to variations in sample restrictions and inclusion of detailed controls for previous scholastic performance or sibling fixed effects. We find that such delays have negative effects on earnings and wages at ages 30 to 40. One additional gap year is associated with 2 percent lower earnings and wages at age 35. This effect is non-trivial in the light of available estimates of the (marginal) returns to schooling in Sweden: according to recent estimates, the wage returns to an additional year of schooling amounts to 5 percent in the year 2000 (le Grand et al, 2001). The effects decline over time and disappear around age 40. We have also examined how gaps affect lifetime earnings and find that two gap years reduce the present value of lifetime earnings to the tune of 40-50 percent of annual earnings at age 40.

We view the negative effects of gaps on earnings and wages as reflecting differential returns to experience before and after university studies. Our estimates suggest that the returns to post-university work experience are substantial whereas the returns to pre-university experience are negligible. However, there is clear evidence of declining marginal returns to post-university work experience, a feature that implies that the earnings profile associated with gap years tends to catch up to the no gap profile. A general lesson is that measures of work experience that do not account for the *timing* of the experience are imperfect indicators of human capital acquired

on the job. Hence, the timing of education should be given a more explicit role in studies of the returns to education.

The analysis of this paper is incomplete as a guide to educational policy. There is a presumption, however, that the social costs to gap years exceed the private costs. The presence of substantial tax wedges is the main reason for this presumption. When designing systems of education finance, such as stipends and subsidized loans, it would therefore seem relevant to recognize how these systems affect the timing of higher education.

References

Albrecht, J, P-A Edin, M Sundström and S Vroman (1999), Career interruptions and subsequent earnings: A reexamination using Swedish data, *Journal of Human Resources* 34, 294-311.

Altonji, J (1993), The demand for and return to education when education outcomes are uncertain, *Journal of Labor Economics* 11, 48-83.

Baum, C (2002), The effects of work interruptions on women's wages, Labour 16, 1-36.

Blinder, A and Y Weiss (1976), Human capital and labor supply: A synthesis, *Journal of Political Economy* 84, 449-472.

Comay, Y, A Melnik and M Pollatschek (1973), The option value of education and the optimal path for investments in human capital, *International Economic Review* 14, 421-435.

Corcoran, M and G Duncan (1979), Work history, labor force attachment, and earnings differences between the sexes, *Journal of Human Resources* 14, 3-20.

Corcoran, M, G Duncan, and M Ponza (1983), A longitudinal analysis of white women's wages, *Journal of Human Resources* 18, 497-520.

Griliches, Z (1980), Schooling interruption, work while in school and the returns from schooling, *Scandinavian Journal of Economics* 80, 294-303.

Gronau, R (1988), Sex-related wage differentials and women's interrupted labor careers: The chicken or the egg, *Journal of Labor Economics* 6, 277-301.

Häkkinen, I (1996), Working while enrolled in university: Does it pay? *Labour Economics* 13, 167-189.

Heckman, J and S Polachek (1974), Empirical evidence on the functional form of the earningsschooling relationship, *Journal of the American Statistical Association* 69, 350-354. Heckman, J, L Lochner and P Todd (2003), Fifty years of Mincer earnings regressions, NBER Working Paper 9732.

Heckman, J, L Lochner and P Todd (2005), Earnings functions, rates of return and treatment effects: The Mincer equation and beyond, NBER Working Paper No. 11544.

Hotz, J, L C Xu, M Tienda and A Ahituv (2002), Are there returns to the wages of young men from working while in school? *Review of Economics and Statistics* 84, 221-236.

Högkoleverket (1997), Tillträde till högre utbildning – en evighetsfråga, Högskoleverkets skriftserie 1997:13 S, Högskoleverket (National Agency for Higher Education).

Landell, E, O Gustafsson and D Grannas (2000), *Utbildningens omvägar* (Detours in Swedish education), ESO-report Ds 2000:58, Fritzes.

Le Grand, C, R Szulkin and M Tåhlin (2001), Lönestrukturens förändring i Sverige (Changes in the Swedish Wage Structure), in *Välfärd och arbete i arbetslöshetens årtionde* (SOU 2001:53), Fritzes.

Light, A (1995), The effects of interrupted schooling on wages, *Journal of Human Resources* 30, 472-502.

Light, A and M Ureta (1995), Early-career work experience and gender wage differentials, *Journal of Labor Economics* 13, 121-154.

Marcus, R (1984), Measuring the rate of return to interrupted schooling, *Journal of Educational Statistics* 9, 295-310.

Mincer, J (1974), Schooling, Experience and Earnings, Columbia University Press.

Mincer, J and S Polachek (1974), Family investments in human capital: Earnings of women, *Journal of Political Economy* 82, S76-S108.

Mincer, J and H Ofek (1982), Interrupted work careers: Depreciation and restoration of human capital, *Journal of Human Resources* 17, 3-24.

Monks, J (1997), The impact of college timing on earnings, *Economics of Education Review* 16, 419-423.

Nordström Skans, O and L Lindqvist (2005), Causal effects of subsidized career breaks, IFAU Working Paper 2005:17.

Nordström Skans, O, Edin P-A and B Holmlund (2006), Wage dispersion between and within plants: Sweden 1985-2000, manuscript IFAU.

Smith, J and R Naylor (2001), Determinants of degree performance in UK universities: A statistical analysis of the 1993 student cohort, *Oxford Bulletin of Economics and Statistics* 63, 29-60.

Stafford, F and M Sundström (1996), Time out for childcare: Signalling and earnings rebound effects for men and women, *Labour* 10, 609-629.

Statistics Sweden (2002), Övergång gymnasieskola – högskola (The transition from upper secondary school to higher education), UF 36 SM 0201.

Statistics Sweden (2003), Universitet och högskolor. Genomströmning och resultat i högskolans grundutbildning t.o.m. 2001/02 (Higher education – throughput and result in undergraduate education up to 2001/02 inclusive), UF 20, SM032.

Statistics Sweden (2004), Övergång gymnasieskola – högskola (The transition from upper secondary school to higher education), UF 36 SM 0401.

Velfærdskommissionen (2005), *Fremtidens velfærd - vores valg*. Report from the Danish welfare commission (<u>http://www.velfaerd.dk</u>).

Appendix

Table A1. Descriptive statistics for cohorts born 1966-1972.

		Men			Women	
	No gap	1-2 yrs gap	3-4 yrs gap	No gap	1-2 yrs gap	3-4 yrs gap
Gap	0.000	1.558	3.358	0.000	1.432	3.409
Estimated Gap	0.071	1.659	3.341	0.084	1.458	3.327
Born in Nordic Countries	0.006	0.005	0.008	0.007	0.006	0.007
Born Western Europe or North America	0.005	0.004	0.003	0.005	0.004	0.003
Other Immigrants	0.019	0.014	0.014	0.023	0.018	0.016
2-year High School	0.047	0.043	0.121	0.059	0.084	0.200
GPA	4.054	3.844	3.481	4.023	3.845	3.581
No father in the data	0.072	0.076	0.080	0.072	0.079	0.083
No mother in the data	0.030	0.029	0.029	0.030	0.031	0.029
Father's Education						
Less than High School	0 248	0.235	0.275	0 248	0 248	0.278
High School	0.183	0.187	0.278	0.230	0.210	0.289
University	0.464	0.466	0.368	0.417	0.414	0.314
Missing	0.404	0.111	0.119	0.105	0.114	0.119
Witssing	0.100	0.111	0.117	0.105	0.114	0.117
Father's Socioeconomic Class						
Nonprofessional Manufacturing	0.035	0.031	0.038	0.041	0.036	0.046
Professional Manufacturing	0.068	0.059	0.078	0.081	0.078	0.098
Service	0.032	0.033	0.044	0.040	0.042	0.052
Low-level White Collar	0.065	0.072	0.077	0.074	0.072	0.081
Intermediate-level White Collar	0.212	0.196	0.221	0.205	0.206	0.210
Higher-level White Collar	0.274	0.272	0.224	0.247	0.238	0.182
Leading position	0.143	0.162	0.134	0.140	0.145	0.135
Farmer	0.020	0.017	0.018	0.025	0.026	0.030
Unclassified Employee	0.026	0.030	0.031	0.027	0.029	0.028
Information missing	0.126	0.128	0.135	0.120	0.129	0.138
Else	0.008	0.006	0.009	0.007	0.008	0.008
Mother's Education						
Less than High School	0.255	0.254	0.284	0.265	0.263	0 207
High School	0.233	0.204	0.204	0.205	0.203	0.297
University	0.485	0.200	0.257	0.255	0.223	0.201
Missing	0.485	0.488	0.400	0.432	0.402	0.372
WIISSING	0.049	0.050	0.055	0.048	0.032	0.049
Mother's Socioeconomic Class						
Nonprofessional Manufacturing	0.015	0.011	0.015	0.018	0.012	0.018
Professional Manufacturing	0.006	0.006	0.006	0.007	0.006	0.009
Service	0.149	0.150	0.197	0.170	0.170	0.213
Low-level White Collar	0.189	0.187	0.200	0.186	0.191	0.207
Intermediate-level White Collar	0.284	0.292	0.284	0.278	0.292	0.270
Higher-level White Collar	0.185	0.185	0.123	0.172	0.160	0.112
Leading position	0.045	0.045	0.045	0.041	0.045	0.040
Farmer	0.011	0.008	0.008	0.012	0.012	0.013
Unclassified Employee	0.021	0.025	0.028	0.022	0.024	0.022
Information missing	0.096	0.093	0.094	0.094	0.088	0.095
Else	0.004	0.004	0.004	0.005	0.003	0.005
Employed at age 30	0.952	0.944	0.936	0.931	0.917	0.900
Number of observations	6 892	15 930	10 574	10 126	20 663	9 373

			0						
		Years				Fraction of time			
	Mean	Sd	Min	Max	Mean	Sd	Min	Max	
Gap	1.72	1.27	0	4					
Duration	4.92	1.20	3	7					
Experience									
Before	0.69	0.89	0	4	0.32	0.31	0	1	
During	1.87	1.50	0	7	0.38	0.28	0	1	
After	8.31	2.31	0	14	0.89	0.18	0	1	
Total	10.87	2.43	0	15	0.72	0.16	0	1	

Table A2. Experience measures at age 35 for cohorts born 1966-67.

Notes: Work experience is for ages 20 to 34. Gaps between high school and university are bounded between 0 and 4 and Duration of university studies are bounded between 3 and 7. Before (during, after) refers to before (during, after) university studies. Fraction of time is the number of years of experience during an interval (before, during or after) divided by the number of years in the interval.

Table A3. Sensitivit	v checks –	variations in	n included	covariates at	age 30 and 35.
	J				

	Earnings			Wages			
	Age 30	Age 30	Age 35	Age 30	Age 30	Age 35	
	(1)	(2)	(3)	(4)	(5)	(6)	
G	-0.034	-0.029	-0.020	-0.042	-0.021	-0.020	
	(0.011)**	(0.003)**	(0.007)**	(0.005)**	(0.001)**	(0.003)**	
Duration	-0.029	-0.034	-0.033	-0.024	-0.022	-0.022	
of studies	(0.011)*	(0.003)**	(0.007)**	(0.005)**	(0.001)**	(0.003)**	
Model							
Sibling fixed	Yes	No	No	Yes	No	No	
effects (#)	(5,199)	110	110	(3,693)	110	110	
Grade	No	Yes	Yes	No	Yes	Yes	
interactions	110		100	110	100	100	
Observations	10,554	70,899	15,209	7,491	57,900	13,060	
Cohorts	1966-72	1966-72	1966-67	1966-72	1966-72	1966-67	

Note: Sibling fixed effects are defined from having the same biological mother, the sibling fixed effects regressions only include individuals with a sibling in the sample. Sibling model also controls for educational fields. Grade interactions model interacts high school grades with both the field of the high school education (60 categories) and the field of the university education (450 categories). Grade interaction model also controls for parent characteristics, county and immigration status. All regressions control for gender and year of birth. Robust standard errors in the parentheses. * significant at 5%; ** significant at 1%.



Figure A1. Description of the structure of the data.

Figure A2. Estimated returns to gap years, by age and cohort.



Note: The figure shows the age pattern of gap estimates based on regressions with log annual earnings as dependent variable. Data for men and women are pooled.

CESifo Working Paper Series

(for full list see www.cesifo-group.de)

- 1728 E. Maarten Bosker, Steven Brakman, Harry Garretsen and Marc Schramm, A Century of Shocks: The Evolution of the German City Size Distribution 1925 1999, May 2006
- 1729 Clive Bell and Hans Gersbach, Growth and Enduring Epidemic Diseases, May 2006
- 1730 W. Bentley MacLeod, Reputations, Relationships and the Enforcement of Incomplete Contracts, May 2006
- 1731 Jan K. Brueckner and Ricardo Flores-Fillol, Airline Schedule Competition: Product-Quality Choice in a Duopoly Model, May 2006
- 1732 Kerstin Bernoth and Guntram B. Wolff, Fool the Markets? Creative Accounting, Fiscal Transparency and Sovereign Risk Premia, May 2006
- 1733 Emmanuelle Auriol and Pierre M. Picard, Government Outsourcing: Public Contracting with Private Monopoly, May 2006
- 1734 Guglielmo Maria Caporale and Luis A. Gil-Alana, Modelling Structural Breaks in the US, UK and Japanese Unemployment Rates, May 2006
- 1735 Emily J. Blanchard, Reevaluating the Role of Trade Agreements: Does Investment Globalization Make the WTO Obsolete?, May 2006
- 1736 Per Engström and Bertil Holmlund, Tax Evasion and Self-Employment in a High-Tax Country: Evidence from Sweden, May 2006
- 1737 Erkki Koskela and Mikko Puhakka, Cycles and Indeterminacy in Overlapping Generations Economies with Stone-Geary Preferences, May 2006
- 1738 Saku Aura and Thomas Davidoff, Supply Constraints and Housing Prices, May 2006
- 1739 Balázs Égert and Ronald MacDonald, Monetary Transmission Mechanism in Transition Economies: Surveying the Surveyable, June 2006
- 1740 Ben J. Heijdra and Ward E. Romp, Ageing and Growth in the Small Open Economy, June 2006
- 1741 Robert Fenge and Volker Meier, Subsidies for Wages and Infrastructure: How to Restrain Undesired Immigration, June 2006
- 1742 Robert S. Chirinko and Debdulal Mallick, The Elasticity of Derived Demand, Factor Substitution and Product Demand: Corrections to Hicks' Formula and Marshall's Four Rules, June 2006

- 1743 Harry P. Bowen, Haris Munandar and Jean-Marie Viaene, Evidence and Implications of Zipf's Law for Integrated Economies, June 2006
- 1744 Markku Lanne and Helmut Luetkepohl, Identifying Monetary Policy Shocks via Changes in Volatility, June 2006
- 1745 Timo Trimborn, Karl-Josef Koch and Thomas M. Steger, Multi-Dimensional Transitional Dynamics: A Simple Numberical Procedure, June 2006
- 1746 Vivek H. Dehejia and Yiagadeesen Samy, Labor Standards and Economic Integration in the European Union: An Empirical Analysis, June 2006
- 1747 Carlo Altavilla and Paul De Grauwe, Forecasting and Combining Competing Models of Exchange Rate Determination, June 2006
- 1748 Olaf Posch and Klaus Waelde, Natural Volatility, Welfare and Taxation, June 2006
- 1749 Christian Holzner, Volker Meier and Martin Werding, Workfare, Monitoring, and Efficiency Wages, June 2006
- 1750 Steven Brakman, Harry Garretsen and Charles van Marrewijk, Agglomeration and Aid, June 2006
- 1751 Robert Fenge and Jakob von Weizsäcker, Mixing Bismarck and Child Pension Systems: An Optimum Taxation Approach, June 2006
- 1752 Helge Berger and Michael Neugart, Labor Courts, Nomination Bias, and Unemployment in Germany, June 2006
- 1753 Chris van Klaveren, Bernard van Praag and Henriette Maassen van den Brink, A Collective Household Model of Time Allocation - a Comparison of Native Dutch and Immigrant Households in the Netherlands, June 2006
- 1754 Marko Koethenbuerger, Ex-Post Redistribution in a Federation: Implications for Corrective Policy, July 2006
- 1755 Axel Dreher, Jan-Egbert Sturm and Heinrich Ursprung, The Impact of Globalization on the Composition of Government Expenditures: Evidence from Panel Data, July 2006
- 1756 Richard Schmidtke, Private Provision of a Complementary Public Good, July 2006
- 1757 J. Atsu Amegashie, Intentions and Social Interactions, July 2006
- 1758 Alessandro Balestrino, Tax Avoidance, Endogenous Social Norms, and the Comparison Income Effect, July 2006
- 1759 Øystein Thøgersen, Intergenerational Risk Sharing by Means of Pay-as-you-go Programs – an Investigation of Alternative Mechanisms, July 2006

- 1760 Pascalis Raimondos-Møller and Alan D. Woodland, Steepest Ascent Tariff Reforms, July 2006
- 1761 Ronald MacDonald and Cezary Wojcik, Catching-up, Inflation Differentials and Credit Booms in a Heterogeneous Monetary Union: Some Implications for EMU and new EU Member States, July 2006
- 1762 Robert Dur, Status-Seeking in Criminal Subcultures and the Double Dividend of Zero-Tolerance, July 2006
- 1763 Christa Hainz, Business Groups in Emerging Markets Financial Control and Sequential Investment, July 2006
- 1764 Didier Laussel and Raymond Riezman, Fixed Transport Costs and International Trade, July 2006
- 1765 Rafael Lalive, How do Extended Benefits Affect Unemployment Duration? A Regression Discontinuity Approach, July 2006
- 1766 Eric Hillebrand, Gunther Schnabl and Yasemin Ulu, Japanese Foreign Exchange Intervention and the Yen/Dollar Exchange Rate: A Simultaneous Equations Approach Using Realized Volatility, July 2006
- 1767 Carsten Hefeker, EMU Enlargement, Policy Uncertainty and Economic Reforms, July 2006
- 1768 Giovanni Facchini and Anna Maria Mayda, Individual Attitudes towards Immigrants: Welfare-State Determinants across Countries, July 2006
- 1769 Maarten Bosker and Harry Garretsen, Geography Rules Too! Economic Development and the Geography of Institutions, July 2006
- 1770 M. Hashem Pesaran and Allan Timmermann, Testing Dependence among Serially Correlated Multi-category Variables, July 2006
- 1771 Juergen von Hagen and Haiping Zhang, Financial Liberalization in a Small Open Economy, August 2006
- 1772 Alessandro Cigno, Is there a Social Security Tax Wedge?, August 2006
- 1773 Peter Egger, Simon Loretz, Michael Pfaffermayr and Hannes Winner, Corporate Taxation and Multinational Activity, August 2006
- 1774 Jeremy S.S. Edwards, Wolfgang Eggert and Alfons J. Weichenrieder, The Measurement of Firm Ownership and its Effect on Managerial Pay, August 2006
- 1775 Scott Alan Carson and Thomas N. Maloney, Living Standards in Black and White: Evidence from the Heights of Ohio Prison Inmates, 1829 – 1913, August 2006

- 1776 Richard Schmidtke, Two-Sided Markets with Pecuniary and Participation Externalities, August 2006
- 1777 Ben J. Heijdra and Jenny E. Ligthart, The Transitional Dynamics of Fiscal Policy in Small Open Economies, August 2006
- 1778 Jay Pil Choi, How Reasonable is the 'Reasonable' Royalty Rate? Damage Rules and Probabilistic Intellectual Property Rights, August 2006
- 1779 Ludger Woessmann, Efficiency and Equity of European Education and Training Policies, August 2006
- 1780 Gregory Ponthiere, Growth, Longevity and Public Policy, August 2006
- 1781 Laszlo Goerke, Corporate and Personal Income Tax Declarations, August 2006
- 1782 Florian Englmaier, Pablo Guillén, Loreto Llorente, Sander Onderstal and Rupert Sausgruber, The Chopstick Auction: A Study of the Exposure Problem in Multi-Unit Auctions, August 2006
- 1783 Adam S. Posen and Daniel Popov Gould, Has EMU had any Impact on the Degree of Wage Restraint?, August 2006
- 1784 Paolo M. Panteghini, A Simple Explanation for the Unfavorable Tax Treatment of Investment Costs, August 2006
- 1785 Alan J. Auerbach, Why have Corporate Tax Revenues Declined? Another Look, August 2006
- 1786 Hideshi Itoh and Hodaka Morita, Formal Contracts, Relational Contracts, and the Holdup Problem, August 2006
- 1787 Rafael Lalive and Alejandra Cattaneo, Social Interactions and Schooling Decisions, August 2006
- 1788 George Kapetanios, M. Hashem Pesaran and Takashi Yamagata, Panels with Nonstationary Multifactor Error Structures, August 2006
- 1789 Torben M. Andersen, Increasing Longevity and Social Security Reforms, August 2006
- 1790 John Whalley, Recent Regional Agreements: Why so many, why so much Variance in Form, why Coming so fast, and where are they Headed?, August 2006
- 1791 Sebastian G. Kessing and Kai A. Konrad, Time Consistency and Bureaucratic Budget Competition, August 2006
- 1792 Bertil Holmlund, Qian Liu and Oskar Nordström Skans, Mind the Gap? Estimating the Effects of Postponing Higher Education, August 2006